Author's response to reviews

Title: Barriers to research utilization and research use among registered nurses in the care of older people: Does the BARRIERS Scale discriminate between research users and non-research users on perceptions of barriers?

Authors:

Anne-Marie Bostrom (anne-marie.bostrom@ki.se)
Kerstin Nilsson Kajermo (kerstin.nilsson.kajermo@ki.se)
Gun Nordstrom (gun.nordstrom@kau.se)
Lars Wallin (lars.wallin@karolinska.se)

Version: 6 Date: 24 April 2008

Author's response to reviews:

Dear Editorial Production Team,

Thanks for editing my article.

1. I have changed all headings and subheadings to “sentence case” (not capital). However, the instrument’s name is spelled with capital litters (BARRIERS Scale).

2. I have a letter from the Journal’s editor that this paper is accepted and I have made the proof-reading. I have got an email that the article (ref 19) will be online at the Journal of Clinical Nursing’s website, until it will be in “paper”.

3 I also prefer tables and figures to fit onto an portrait A4 page, however, I had to add four new columns with the t-values and r-values to Table 2. These new columns made it impossible to have Table 2 to fit onto portrait page. If you have the skills and knowledge to change this format, please do so. I don’t have these skills and knowledge.

Best regards,

Anne-Marie Boström